Author's response to reviews

Title: The French national prospective cohort of patients co-infected with HIV and HCV (ANRS CO13 HEPAVIH): Early findings, 2006-2010

Authors:

Marc-Arthur Loko (Marc-Arthur.Loko@isped.u-bordeaux2.fr)
Dominique Salmon (dominique.salmon@cch.aphp.fr)
Patrizia Carrieri (Patrizia.Carrieri@aol.com)
Maria Winnock (maria.winnock@gmail.com)
Marion Mora (marion.mora@inserm.fr)
Laurence Merchadou (Laurence.Merchadou@isped.u-bordeaux2.fr)
Stéphanie Gillet (stephanie.gillet@isped.u-bordeaux2.fr)
Elodie Pambrun (Elodie.pambrun@isped.u-bordeaux2.fr)
Jean Delaune (jean.delaune@isped.u-bordeaux2.fr)
Marc-Antoine Valantin (marc-antoine.valantin@psl.ap-hop-paris.fr)
Isabelle Poizot Martin (isabelle.poizot@mail.ap-hp.fr)
Didier Neau (didier.neau@chu-bordeaux.fr)
Philippe Bonnard (philippe.bonnard@tnn.aphp.fr)
Eric Rosenthal (rosenthal.e@chu-nice.fr)
Karl Barange (barange.k@chu-toulouse.fr)
Philippe Morlat (philippe.morlat@chu-bordeaux.fr)
Karine Lacombe (karine.lacombe@sat.aphp.fr)
Anne Gervais (anne.gervais@bch.aphp.fr)
François Rouges (francois.rouges@avc.aphp.fr)
Alain Bicart See (abicart@hjd.asso.fr)
Caroline Lascoux-Combe (caroline.lascoux-combe@sls.aphp.fr)
Daniel Vittecoq (daniel.vittecoq@pbr.aphp.fr)
Cecile Goujard (cecile.goujard@bct.aphp.fr)
Claudine Duvivier (duvivier@pasteur.fr)
Bruno Spire (bruno.spire@inserm.fr)
Jacques Izopet (Izopet.j@chu-toulouse.fr)
Philippe Sogni (philippe.sogni@cch.aphp.fr)
Lawrence Serfaty (lawrence.serfaty@sat.aphp.fr)
Yves Benhamou (ybenhamou@teaser.fr)
Firouze Bani-Sadr (firouze.bani-sadr@tnn.aphp.fr)
François Dabis (francois.dabis@gmail.com)

Version: 2 Date: 26 July 2010

Author's response to reviews: see over
Dear Sir,

According to your request, we have included in our manuscript entitled "The French national prospective cohort of patients co-infected with HIV and HCV (ANRS CO13 HEPAVIH): Early findings, 2006-2010":

- an abstract with the correct format
- a title page at the front of the manuscript file including the names of the authors, their affiliation and e-mail addresses as well as the full postal address of the submitting author
- a paragraph in the method section stating that the research conformed to the Helsinki Declaration and the name of the ethics committee which approved the research
- an author's contribution section after the discussion section
- a declaration of competing interests after the conclusion section

Lastly figure 3 was completed by adding confidence intervals.

Looking forward to hearing from you,

Yours sincerely,